Author's response to reviews

Title: The Multi-Center Canadian Acellular dermal matrix Trial (The MCCAT Study): Study protocol for a randomized controlled trial in implant-based breast reconstruction

Authors:

Toni Zhong (toni.zhong@uhn.ca)
Claire Temple Oberle (Claire.Temple-Oberle@albertahealthservices.ca)
Stefan Hofer (stefan.hofer@uhn.ca)
Brett Beber (b.beber@utoronto.ca)
John Semple (john.seemple@wchospital.ca)
Mitchell Brown (Mitchell.brown@wchospital.ca)
Sheina Macadam (drsmacadam@gmail.com)
Peter Lennox (klennox@telus.net)
Tony Panzarella (panzar@uhnresearch.ca)
Colleen McCarthy (McCarthC@mskcc.org)
Nancy Baxter (BaxterN@smh.ca)

Version: 2 Date: 9 September 2013

Author's response to reviews:

Please see the uploaded cover letter. Thank-you.